Author's response to reviews

Title: Evaluation of Stepping Stones as a tool for changing knowledge, attitudes and behaviors associated with gender, relationships and HIV risk in Karnataka, India

Authors:

Janet Bradley (jbradley@uresp.ulaval.ca)
Parinita Bhattacharjee (parinita@khpt.org)
BM Ramesh (bmramesh@khpt.org)
M Girish (megirish2001@gmail.com)
Arup Kumar Das (arup@khpt.org)

Version: 13 Date: 2 June 2011

Author's response to reviews:

The reviewer noted a few typos only. I think these have all now been corrected.